Author's response to reviews

Title: Advertising and disclosure of funding on patient organisation websites: a cross-sectional survey

Authors:

Douglas E Ball (douglasball@yahoo.co.uk)
Klara Tisocki (ktisocki@yahoo.ie)
Andrew Herxheimer (a@herxheimer.net)

Version: 4 Date: 1 August 2006

Author's response to reviews:

Latest revisions in line with request by editorial team:

1. Website addresses on page 18 removed and moved to references (as new reference 33)

2. Reference 14 was changed to reference 17 in the latest revision with subsequent changes to other reference numbers. These changes were required due to moving websites from the text to the reference list.

3. Author contributions changed to Authors’ contributions

4. Lines have been added to Table 1 to define the borders of the table.

Regards
Douglas Ball